

### Dynamic modeling of global fossil fuel infrastructure and materials needs: Overcoming a lack of available data

Hugo Le Boulzec, Louis Delannoy, Baptiste Andrieu, François Verzier, Olivier

Vidal, Sandrine Mathy

### ▶ To cite this version:

Hugo Le Boulzec, Louis Delannoy, Baptiste Andrieu, François Verzier, Olivier Vidal, et al.. Dynamic modeling of global fossil fuel infrastructure and materials needs: Overcoming a lack of available data. Applied Energy, 2022, 326, 10.1016/j.apenergy.2022.119871. hal-03780879

### HAL Id: hal-03780879 https://hal.science/hal-03780879v1

Submitted on 19 Sep 2022

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

### Highlights

# Dynamic modeling of global fossil fuel infrastructure and materials needs: overcoming a lack of available data

Hugo Le Boulzec, Louis Delannoy, Baptiste Andrieu, François Verzier, Olivier Vidal, Sandrine Mathy

- Fossil fuel infrastructures are modeled from 1950 to 2050.
- A dynamic material flow analysis (MFA) is developed and applied.
- Embedded materials, energy and CO<sub>2</sub> emissions are estimated.
- A potential to build renewable technologies from fossil fuel infrastructures is identified.
- To be ambitious, decommissioning strategies should enhance in-use steel recycling.

### Dynamic modeling of global fossil fuel infrastructure and materials needs: overcoming a lack of available data

Hugo Le Boulzec<sup>*a,e,\**</sup>, Louis Delannoy<sup>*b,c*</sup>, Baptiste Andrieu<sup>*a,d*</sup>, François Verzier<sup>*a*</sup>, Olivier Vidal<sup>*a*</sup> and Sandrine Mathy<sup>e</sup>

<sup>a</sup>Univ. Grenoble Alpes, Univ. Savoie Mont Blanc, CNRS, IRD, Univ. Gustave Eiffel, ISTerre, 38000 Grenoble, France

<sup>b</sup>Univ. Grenoble Alpes, CNRS, Inria, LJK, STEEP 38000 Grenoble, France

<sup>c</sup>Petroleum Analysis Centre, Staball Hill, Ballydehob, West Cork, Ireland

<sup>d</sup>The Shift Project, 16-18, rue de Budapest - 75009 Paris, France

<sup>e</sup>Univ. Grenoble Alpes, CNRS, INRAE, Grenoble INP, Grenoble Applied Economics Laboratory (GAEL), 38000 Grenoble, France

#### ARTICLE INFO

Materials requirements Embedded energy

Embedded CO<sub>2</sub> emissions

Keywords:

Fossil fuels

Energy transition

#### Abstract

The low-carbon energy transition requires a widespread change in global energy infrastructures which in turn calls for important inputs of energy and materials. While the transport and electricity sectors have been thoroughly analyzed in this regard, that of the hydrocarbon industry has not received the same attention, maybe in part due to the difficulty of access to the necessary data. To fill this gap, we assemble public-domain data from a wide variety of sources to present a stock-flow dynamic model of the fossil fuels supply chain. It is conducted from 1950 to 2050 and along scenarios from the International Energy Agency. We estimate the concrete, steel, aluminum and copper requirements for each segment, as well as the embedded energy and CO2 emissions through a dynamic material flow analysis (MFA) model. We find that (i) the material intensities of oil, gas and coal supply chains have stagnated for more than 30 years; (ii) gas is the main driver of current and future material consumption; and (iii) recycled steel from decommissioned fossil fuels infrastructures could meet the cumulative need of future low-carbon technologies and reduce its energy and environmental toll. Furthermore, we highlight that regional decommissioning strategies significantly affect the potential of material recycling and reuse. In this context, ambitious decommissioning strategies could drive a symbolic move to build future renewable technologies from past fossil fuel structures.

#### 1. Introduction

Human society and nature interactions have experienced a radical change over the 20<sup>th</sup> century, driven by the rapid industrialization of Western economies and that of several emerging countries in more recent years [1, 2, 3]. Economic expansion is largely attributable to the availability and affordability of fossil resources, whose high energy return on investment (EROI) have triggered productivity gains [4, 5]. The abusive use of energy has however unleashed unprecedented climate and ecological damages, jeopardizing the planet's life supporting functions [6].

As a tentative response, 196 countries gathered during the 2015 United Nations Climate Change Conference of Paris and agreed to hold the increase in the global average temperature to well below 2°C above pre-industrial levels. Such a target requires deep transformations of our society including, but not limited to, a shift away from fossil fuels to low-carbon energies, which calls for important quantities of energy and materials [7, 8, 9, 10, 11, 12]. While the transport and electricity sectors have been thoroughly analyzed, the hydrocarbons have not received the same attention, to the best of the authors' knowledge<sup>1</sup>. This may in part be due to the fact that infrastructure data for the fossil fuel industries are relatively difficult to access, and are often commercially

sensitive. This makes the analysis of this sector particularly complicated for the scientific community not affiliated with fossil fuel producers or operators. To fill this gap, we present a stock-flow dynamic modeling of the infrastructures as a part of the fossil fuels supply chain, from 1950 to 2050 and along scenarios from the International Energy Agency (IEA). We estimate the concrete, steel, aluminum and copper requirements for each segment, as well as the embedded energy and CO<sub>2</sub> emissions through a dynamic material flow analysis (MFA). Finally, the potential of recycling steel from decommissioned fossil fuel infrastructures to build power technologies is further discussed.

This paper is structured as follows. Section 2 examines the existing literature. Section 3 describes the methodology and data used. Section 4 presents the results, section 5 discusses them, section 6 details the robustness and limitations of the study and section 7 concludes.

#### 2. Literature review

#### 2.1. An evolving field

Modeling the stocks and flows of metals or minerals that make the infrastructures has been a research topic for more than twenty years. Early publications focus on static global or regional MFA of the base materials used over the 20<sup>th</sup> century-such as copper and steel. The Stocks and Flows Project (STAF) launched in 2000 by the Center for Industrial Ecology is one example [13, 14, 15, 16]. These studies allow for top-down analysis, relying on material databases

<sup>\*</sup>Corresponding author: hugo.le-boulzec@outlook.fr

ORCID(s): 0000-0003-3658-7715 (H. Le Boulzec); 0000-0002-5821-2597 (L. Delannoy); 0000-0002-1485-3165 (B. Andrieu); 0000-0002-7998-4115 (S. Mathy)

<sup>&</sup>lt;sup>1</sup>The relevant literature is reviewed in Section 2.

such as the Internation Copper Study Group [17] and the World Bureau of Metal Statistics [18]<sup>2</sup>. The authors aim at providing insights in the material cycles within human society, and quantifying the losses-dissipation, landfills-and the potential of secondary use. Materials analysis has progressively evolved towards dynamic studies, and bottom-up approaches<sup>3</sup> enable a better understanding of the technological drivers of raw materials demand over time. A significant number of these publications deal with the building sector because of waste generation and secondary resources aspects [19, 20, 21], stock drivers [22], the effect of lifetime [23] or materials demand in transition scenarios [24, 25]. More recently, several authors also analyze the material facet of the energy sector with growing consideration on the consequences of the ongoing energy transition.

#### 2.2. The energy transition challenge

The transition requires vast amounts of minerals and metals [9, 26, 27] due to a significantly higher material intensities of renewables [28, 29, 8] and new technologies [30]. Vidal et al. [8] state that "a shift to renewable energy will replace one non-renewable resource (fossil fuels) with another (metals and minerals)", while Li et al. [31] point out the trend toward a "more metal-intensive energy future" as renewable energy are increasingly being developed. Recent dynamic MFA or simple flow analysis<sup>4</sup> focus on quantifying the prospective demand in non-energetic materials for a low-carbon energy system in various institutional scenarios (e.g. IEA-450, IEA Bluemap, NDRC scenarios), nongovernmental scenarios (e.g. Greenpeace, WWF-Ecofys) or academic scenarios (e.g. SSP). Researchers show that the most environmentally ambitious scenarios display the highest material constraints [34, 35], and identify two areas of concern: criticality, which is defined as the levels of importance in use and availability of a material [36], as well as environmental and energy impacts.

Among low-carbon technologies, the most significant material constraint could stem from solar PV and wind turbines along with power storage [37, 35] and the transmission and distribution networks [38]. A higher level of pressure for rare-earth elements (REE) than for base metals is emphasized, with potential constraint in REE for each PV solar technology [29, 39, 40], region-specific supply risk [41, 42, 43, 44, 45] which require future increase in numerous metal supplies [46, 47] with potential limitations in reserves which could hamper future projects [48]. The downscaling

of a global issue to a political scale also paves the way to the criticality literature, estimating potential supply risks for various situations [49, 50, 51, 52, 53, 54, 55]. Future constraint on raw materials also comes from the energy and environmental impact of their production. While the criticality research field focuses on technological materials, impact analyses are mainly conducted on base materials though the link between energy and materials.

#### 2.3. Recycling and the energy-raw materials nexus

The energy-material nexus remains insufficiently addressed in energy transition scenarios, and several authors call for a better understanding of its impact [34, 35] and for a better understanding of the assessment of the emission costs and gains of the energy transition [31]. Elshkaki [39] further estimates that the global energy demand of material production for photovoltaic solar technologies could represent up to 12% of solar power generation in the IEA-450 scenario by 2050, and its CO<sub>2</sub> emissions could reach up to 2.2% of the global emissions. Several policy tools are incorporated in the publications, in order to discuss the upcoming challenges. First, recycling could diversify supply sources to reduce pressure on primary materials [35, 56] for geopolitical purposes [46], and environmental concerns as dilution of metals in deposits rises [57, 58]. However, the short-term objective of energy transition is equivalent to lifetimes of technologies, thus limiting the potential of recycling before 2050 [9, 59, 45]. Second, studies emphasize that material supply pressure could be mitigated by reducing the material content in power generation technologies [40, 34].

#### 2.4. Toward more systematic analysis

The upcoming material challenge is also addressed in integrated assessment models (IAMs), as a necessary step to produce "biophysically more consistent descriptions of society's future metabolism" [60]. IAMs allow to quantify the material requirements in scenarios of low-carbon transitions from a system perspective. For instance, Hache et al. [61, 62] and Seck et al. [63] use a modified version of the TIMES model (TIAM-IFPEN) to study the critical materials for the transportation sector electrification, the impact of the future power generation on cement demand and the copper availability up to 2050. Sverdrup et al. [64, 65] and Olafsdottir and Sverdrup [66] focus on nickel, helium, copper, zinc and lead using the WORLD7 model. Using on the MEDEAS integrated assessment model, Capellàn-Pérez et al. [67] assess the material investments necessary for a transition to renewable energies, in a Green Growth narrative, concluding it to be unrealistic.

The literature on the metals and minerals supply pressure stemming from the energy transition has been recently booming. However, despite the large share of fossil fuels in the global primary energy consumption, researchers have to date and to the best of the authors' knowledge not explored in detail the material requirements of the industry on a global

 $<sup>^2\</sup>mathrm{A}$  top-down study is defined as a modeling relying on input output tables which assesses stocks from a flow analysis.

<sup>&</sup>lt;sup>3</sup>Bottom-up modeling focuses on describing parts of the system in details and then conducting a part-whole modeling.

<sup>&</sup>lt;sup>4</sup>Further modeling on the topic has been conducted through Life-Cyle Analyses. For example, Kleijn et al. [28] examine the metals demand and  $CO_2$  emissions of low-carbon energies for the power sector, in regard to four different scenarios, while Koning et al. [32] estimate the required extraction of 11 metals until 2050 under technology-specific low-carbon scenarios, for the electricity and road transportation systems and Hertwitch et al. [33] take on the issue by approaching the needs in steel, cement, copper and aluminum for power generation technologies in the IEA BLUE Map and Baseline scenarios.

scale and in a long-term perspective<sup>5</sup>. This study attempts to tackle this question and fill the relative literature gap that exists today. To do so, the following methodology is carried out.

#### 3. Methods and data

#### 3.1. The DyMEMDS model

The *DyMEMDS* (Dynamic Modelling of Energy and Matter Demand and Supply) model is developed to estimate the needs in primary and recycled materials, as well as the energy and  $CO_2$  emissions resulting from the extraction of these materials, along global or national energy scenarios. The entire energy chain is covered from primary energy to final consumption with forty technologies of energy production, storage, transport, transformation along three sectors of energy use: transport, construction and industry [58]. A multi-sector analysis of material stocks and flows is prepared for further work. Only the fossil fuels supply chain module is described in this section<sup>6</sup>.

Historical infrastructures data and energy scenarios are used to model the hydrocarbon supply chain. When no data are found, we use Gompertz functions of Gross Domestic Product (GDP) and population or correlations with the energy supply. Technological material intensities, recycling rates, normal lifetime distributions and end-of-life recycling rate (EOL-RR) are used to estimate retrospective and prospective flows and stocks of materials. Indirect impacts of the materials demand are evaluated relying on state-ofthe-art works on embodied energy and  $CO_2$  emissions of materials. A local sensitivity analysis is finally conducted to assess the effects of the mean lifetime and materials intensities uncertainties. Figure 1 presents the methodology lying behind the *DyMEMDS* model.

The most important box, 'raw materials stocks and flows', can be extended for further clarification. The primary production flow in tons/year feeds an in-use stock in tons of materials immobilized in the infrastructure. At the end of the lifetime of the infrastructure considered, the in-use stock is reduced by the flow of materials to be recycled. Normal lifetime distributions are considered in this study. The losses flow during the recycling process thus feeds the cumulative losses stock, while the end-of-life recycling flow builds a recycled stock. The latter then allows for secondary production of a recycled materials flow, which is ultimately reused in the *in-use stock*. New infrastructures thus mobilize primary and secondary production. In the specific case of a reduction of the infrastructure stock, there is a dismantling before the end of the life span. An unused flow is then created, ultimately feeding an unused stock, which then follows a recycling path. This stock could eventually produce a reused flow for other infrastructures. The infrastructures embodied energy and CO2 emissions are finally estimated through assumptions about the material production energies

and  $CO_2$  emissions. Figure 2 details the key loop used in the *DyMEMDS* model to estimate the stocks and flows of materials during the construction and maintenance of the infrastructure.

#### 3.1.1. Infrastructures

The infrastructures considered in this study are separated into three main segments according to the historical distinction made in the petroleum industry: upstream, midstream and downstream activities. It totals 11 technologies used in supply activities from exploration to distribution. The fossil fuels supply chain encompasses a larger number of infrastructures some are excluded due to a lack of data (oil tanks, national coal transport and processing), large disparities in the data obtained (drilling rigs)<sup>7</sup>, or difficulties to model future distribution in technologies related to decentralized demand (petrol stations and compressed natural gas services). Figure 3 displays the infrastructures typology used in *DyMEMDS*.

#### 3.1.2. Dynamic evolution of infrastructures

A stock-driven approach is conducted in order to model the dynamic evolution of fossil fuels associated infrastructures. The evolution of technology stocks is assumed to be correlated with the fossil fuels production (eq 1)<sup>8</sup>. Infrastructures are then attributed to a fossil fuel, either fully (eg. oil tankers only carry petroleum products) or in a proportionate manner (eg. wells produce both oil and gas). In this case, the evolution of each infrastructure stock is correlated to its related share of fossil products in the supply chain using historical and prospective shares of oil, gas and coal traded from BP [71] and the IEA<sup>9</sup> [72].

Substantial annual variations are observed in hydrocarbon production, but only wells experience significant upwards and downwards annual changes. An adapted function of each fossil fuel supply was established to prevent these variations. It is defined as a third-order exponential moving average<sup>10</sup> (eq 2), which enables both to smooth out the short-term impact of price variations on production and to overweight the more recent values.

$$S_{t,i} = \alpha_i \cdot \text{EMA}_{3,P_i}(t) \tag{1}$$

$$\text{EMA}_{3,P_i}(t) = P_i(t) \cdot 0.5 + \text{EMA}_{3,P_i}(t-1) \cdot 0.5$$
 (2)

<sup>&</sup>lt;sup>5</sup>Only Wang et al. [68] proposed an estimation of the steel weight of the drilled wells and the transportation pipelines in China.

 $<sup>^{6}\</sup>mathrm{A}$  detailed description of the inputs is provided in the Supplementary Information.

<sup>&</sup>lt;sup>7</sup>Significant disparities were observed in both material intensities and rigs census. It is explained by both the recent and rapid development of floating structures and the wide range of structure sizes.

<sup>&</sup>lt;sup>8</sup>Some researchers rely on logistic or Gompertz functions, see for instance the work of Gutiérrez et al. [69] for vehicle stocks and Huang et al. [70] for buildings stocks. The link between hydrocarbon technologies and energy prices, as well as political decisions (e.g. OPEC's influence on prices) hamper the application of this methodology here.

<sup>&</sup>lt;sup>9</sup>A high share of traded oil is noticed, while coal and natural gas are mostly consumed locally.

<sup>&</sup>lt;sup>10</sup>A moving average is defined as an average of a predefined number of past values in a dataset, which are weighted equally in its simple form. In its exponential form, the weight of the values decreases exponentially, which enables to outweigh the more recent data.



Figure 1: Simplified methodology lying behind the DyMEMDS model.



Figure 2: Key loop used in the *DyMEMDS* model to estimate the stocks and flows of materials during the construction and maintenance of the infrastructure.

With  $S_{t,i}$  the stock of infrastructure i in the year t,  $\alpha_i$  the correlation coefficient for the infrastructure i and EMA<sub>3,P<sub>i</sub></sub>(t) the third-order exponential moving average in year t of the hydrocarbon supply  $P_i$  related to the infrastructure i.

#### 3.1.3. Material recycling

This study focuses on four bulk materials, namely aluminum, concrete, copper and steel, which represent a significant share of material consumption [73], energy demand and environmental impacts of material production [74]. No material substitution is assumed for the base materials, but some studies present contrasted views, e.g. Sverdrup et al. [75, 76] A dynamic end-of-life recycling rate (EOL-RR) is assumed for each material, encompassing both the recycling process efficiency rate (EOL-PR) and the old scrap collection rate (EOL-CR), as defined by Glöser et al. [77] and Graedel et al. [78]. A global and multi-sectoral averaged EOL-CR is assumed in this study. Table 1 summarizes the considered rates.

	1050	2010	2030	2050
	1950	2010	2030	2030
Concrete	0	0	0	0
Steel	0.6	0.7	0.74	0.84
Aluminum	0.39	0.45	0.48	0.54
Copper	0.34	0.4	0.42	0.48

#### Table 1

End-of-life recycling rates of the considered materials.



Figure 3: Oil, gas and coal infrastructures from production to final users supply. The infrastructures in grey and italic are not considered in this study.

#### 3.2. Materials, energy and CO<sub>2</sub> intensities

The material requirements for the hydrocarbon production infrastructures are obtained from Monfort et al. [79]. The dataset extensively relies on the Ecoinvent database<sup>11</sup>, which provides with useful data for a large panel of infrastructures. Remaining intensities are obtained from both academic publications and private firms reports. The material intensity data used in this study come mostly from sources dating from the 1960s to the 2000s. The correlation between infrastructure stock and hydrocarbon production justifies the temporal validity of these data for the study, since nearly 60% of historical oil production and 57% of historical gas production took place between 1960 and 2000 [80]. The embodied energies of primary and secondary materials are dynamic between 1900 and 2100, and their scope extends from cradle to gate. Primary metals embodied energies depend on both the ore grade of the mined material and the technological improvement of production technologies [81, 82, 83]. Data from Vidal [84] and Vidal et al. [85] are considered for copper, aluminum, and steel (assuming all steel alloys represent steel). The production energy of concrete is estimated from a multi-regional approach to cement production and clinker energy intensities. Energy of metal recycling is assumed to be consumed during secondary production as a fraction of the energy of primary production. An annual technological progress in production processes of 1% is estimated between 1900 and 2100, as well as a floor value equal that of the thermodynamic limit of the recycling processes. The minimum energy of recycling equals the melting energy of each metal plus the energy of collection and sorting. CO<sub>2</sub> intensities of materials production are also considered in the model:

they are dynamic and vary according to the  $CO_2$  intensity of the energy consumed over time. We consider a decreasing intensity since 1900, and assume that 83g of  $CO_2$  are emitted by MJ of consumed energy in 2015. This value is supposed constant from 2015 onward.

#### 3.3. Scenarios

Two scenarios from the International Energy Agency (IEA) are selected: the Stated Policies Scenario (STEPS) and the Net Zero Emissions by 2050 (NZE) [86, 87].

- The STEPS considers government measures already in place or officially announced and envisions a quick return to pre-COVID-19 consumption levels. By 2050, the electricity mix is 55% renewable, coal consumption falls by 15% from 2020, oil consumption rises by 15% and gas consumption rises sharply by 50%. The effects of climate change are limited to an average global warming of 2.7°C in 2100 compared to preindustrial levels (Fig. 4).
- The NZE scenario offers a more ambitious approach. Considered by the IEA as a deepening of the former Sustainable Development Scenario (SDS) [88], it assumes the achievement of a carbon neutrality of the energy sector by 2050, which allows to limit the average global warming to 1.5°C. Renewable energies play a predominant role, with solar and wind power accounting for nearly 70% of electricity production in 2050. Moreover, energy demand falls by 7% between 2020 and 2050, with a decline in coal consumption of 90%, 75% for oil and 55% for gas over the same period (Fig. 4). When some information is missing, data from the Sustainable Policies (SP) scenarios are used instead [72].

<sup>&</sup>lt;sup>11</sup>The Ecoinvent database contains life-cycle inventory data for a wide range of technologies, with a thorough documentation on their hypothesis and validity (e.g. geographical and temporal).



Figure 4: Historical and prospective global primary energy supply from fossil fuels in the STEPS and NZE scenarios.

#### 3.4. Sensitivity analysis

A local sensitivity analysis was conducted to assess the impact of the input parameters uncertainties on outputs. Previous studies showed the significant share of lifetime uncertainty impacts in sensitivity analysis [23, 89], and further pointed out that the standard deviation and shape of lifetime distributions had little effects on the results compared to the mean value [90, 77, 91]. The sensitivity analysis therefore focuses on the mean lifetime value in the normal distribution, for which variations of  $\pm 20\%$  were computed.

#### 4. Results

#### 4.1. Infrastructures

#### 4.1.1. Gas

The gas infrastructures experience a significant increase in the STEPS and a drop after a peak in 2030 in the NZE scenario. The largest increase in infrastructure occurs for gas in the STEPS. The world fleet of LNG carriers reaches more than 1200 units in 2050 in the STEPS, the storage capacity exceeds 136 million cubic meters and the total capacity of liquefaction and regasification plants doubles to more than 3700 billion cubic meters between 2020 and 2050. Conversely, the carbon neutral NZE scenario involves 40% decrease in total LNG supply chain capacity between 2020 and 2050. The number of LNG carriers drops to 376, after peaking at 788 vessels in 2030. The global pipeline network displays a similar trend, raising to 15 million kilometers in 2050 in the STEPS, and dropping to less than 5 million kilometers in the NZE scenario (Fig. 5). The modeled values of the infrastructures stocks are available in the Supplementary Information.

#### 4.1.2. Oil

Two trends of the infrastructure of the oil supply chain can be identified. After a slight increase between 2020 and 2030, the stock stabilizes in 2030 in the STEPS, while a slowdown is observed in the NZE scenario between 2020 and 2025 and followed by a substantial reduction. This unveils a significant constraint on the infrastructures mobilized in the oil supply stages. Pipelines, tankers and refineries experience a drop in their stock by nearly 75% between 2020 and 2050. The number of pipelines decreases to 242,000 km, and the tanker fleet to only 150 Million Dead Weight Tons (MDWT)<sup>12</sup>. In the STEPS, the growth of the traded oil shares combined with the rise of the oil consumption induces a faster development of transport infrastructures than storage or refineries. The stock of petroleum products pipelines reaches 1.08 million kilometers in 2050, and the fleet of tankers 670 MDWT, while the global refining capacity increases to nearly 108 million barrels per day. Wells development depends the cumulative consumption. The number of wells thus slightly raises from 1.35 million to 1.4 million in 2050 in the STEPS and experiences a substantial decrease to 300,000 in the NZE scenario (Fig. 5).

#### 4.1.3. Coal

The two global scenarios consider a drop in coal consumption between 2020 and 2050, albeit of varying magnitude. The bulk carrier fleet thus collapses over the period and finally reaches 150 MDWT and 21 MDWT in 2050, respectively, in the STEPS and NZE scenarios (Fig. 5).

#### 4.2. Materials

Steel is the main source of materials requirements for fossil fuels. It represented nearly 70% of the sector materials demand between 1950 and 2020, and the remaining 30% is mainly concrete. In contrast, aluminum and copper consumption represent 0.04% and 0.06%, respectively. However, these proportions conceal major disparities in absolute annual consumption, depending on the evolution of the oil, gas and coal markets. A first peak in consumption can be observed in 1973, following the increase in hydrocarbon consumption since the early 1940s. The annual demand for concrete raised from 2.8 Mt/yr to 42 Mt/yr and steel consumption grown from 4.4 Mt/yr to 64 Mt/yr between 1940 and 1973. The succession of two oil peaks in 1973 and 1979 then reduced the demand for materials to almost zero for steel, concrete and copper, while the dependency of aluminum demand to oil products infrastructures drove its value to zero. The knowledge of historical material demand

<sup>&</sup>lt;sup>12</sup>The DWT is a unit of the total carriage capacity of a vessel, including all types of goods and passengers.



Figure 5: Historical and prospective stocks of key infrastructures in the STEPS and NZE scenarios, global scale.



Figure 6: Historical material intensities in the fossils supply sector from 1950 to 2016.

allows to estimate historical materials intensities for the global fossils supply chain. Their values between 1950 and 2016 are displayed in Fig. 6. We observe a stabilization of their level since the 1970s, despite a significant drop in the sector's aluminum intensities in the 1990s.

#### 4.2.1. Material consumption in the scenarios

The historical distribution pattern of the cumulative consumption among the four structural materials remains similar in the scenarios. Steel accounts for 70% of the cumulative sectoral material demand between 2020 and 2050 for the STEPS and over 77% in the NZE scenario. The share of concrete in the total consumption experiences a decrease in

the scenarios considered and aluminum and copper shares remain low. The increase of the hydrocarbons demand until 2025 induces higher peak material requirements than those observed historically. The steel, aluminum and copper demands respectively reach 96 Mt, 80 kt and 0.44 Mt in 2025. In the STEPS, the cumulative flow of steel needed between 2020 and 2050 is 2.4 Gt and that of concrete exceeds 1 Gt. This represents 1.4 years of total global production at 2020 levels for steel, and a small amount compared to an estimated total global production of 20 to 24 Gt in 2020 for concrete [92, 93]. The NZE scenario requirements reach 0.34 Gt for steel and 0.1 Gt for concrete. An identical trend is observed for copper and aluminum. The cumulative demand for copper reaches 11 Mt in the STEPS, compared to a total global production of 20 Mt in 2020, and cumulative demand drops to only 1.5 Mt in the NZE scenario. The total aluminum requirements amount to 0.86 Mt for the STEPS and 51 kt for the NZE, which respectively represents 1.3% and 0.08% of 2020 total global production. The combination of this lower consumption and the identical historical stock allows the NZE scenario to rely more heavily on recycled materials. The proportion of recycled metals reaches more than 80% in the NZE scenario, and 29% to 42% in the STEPS scenario. The stock of concrete reaches more than 1.2 billion tons in 2050, while a substantial drop to 0.35 billion tons is observed in NZE scenario. This trend is similar for the three structural metals, and steel stock amounts to 2.9 billion tons in the STEPS and 0.82 billion tons in the NZE scenario in 2050 (Fig. 7). The peaks and drops observed in the stock evolution in 2017 are explained by the differences between the historical and prospective data.

# 4.2.2. Material consumption per fossil fuels and segments

The oil and gas infrastructures represent most of the raw materials, which is explained by the complexity and decentralization of the supply infrastructures. The aluminum is entirely consumed by the oil product pipelines for all scenarios-due to underlying data-and the midstream segment therefore dominates aluminum consumption. Conversely, the split between oil and gas remains more equal for concrete, steel and copper, although there is a clear trend towards an increase in the share of gas in materials demand over time. The share of gas in concrete consumption evolves from 30% to nearly 60% between 1950 and 2050 and from 45% to nearly 70% over the same period for steel. The distribution between the segments also shows important differences. The upstream segment represents most of the concrete demand between 1950 and 2050-in the production of wells-while a more equal distribution is observed for steel and copper.

#### 4.2.3. Material consumption per infrastructures

The bulk materials demands show high level of domination of some infrastructures. Wells represented about 80% of concrete consumption until 1970 and their share declined with the progressive development of gas pipelines, which share reaches about 25% in 2050 in the STEPS. A slight increase of the oil pipeline share was noticed, finally reaching 5% in 2050. The gas network also carries most of the sectoral steel demand. Its share experienced a rise from 45% of the consumption between 1950 and 1970 to almost 60% in 2050 in the STEPS. The wells have an opposite profile and the oil pipelines represent 20% of steel demand over the period. Copper is mostly consumed in steel alloys in the hydrocarbon supply chains. It therefore displays a distribution profile similar to steel. Finally, nearly all aluminum is consumed in oil pipelines-due to the underlying data.

## 4.2.4. Opportunity and uncertainties in the future amount of recycled materials

The share of recycled materials in the hydrocarbon supply chain increased over time. It evolved from 3% in 1950 to nearly 20% in 2020, with disparities depending on the metals considered. A future stabilization or drop in the fossil fuels demand could further induce a substantial rise of the recycled share of metals inflow in 2050, as pointed out in the STEPS and NZE scenarios. The combination of this trend with a lower production energy of recycled metals results in an overall decline of the embodied energy and CO<sub>2</sub> emissions in the oil, gas and coal infrastructures. A high recycling content of material depends on the EOL-RR of the hydrocarbon supply industry, which could be hampered by the evolution of the EOL-CR of the fossil fuels supply sector. The future of the unused infrastructures is a growing research field (e.g. OGUK [94] in the United Kingdom and Kaiser [95] in the Gulf of Mexico), and regional political decommissioning strategies will have significant impacts on the EOL-CR. Simulation of several decommissioning strategies by declines of 20% and 50% of the EOL-CR on the obsolete stock of steel are displayed in Figure 8. An estimated 0.8 billion tons of steel alloys are estimated to be accumulated in landfills or leave in-situ by 2050 in the STEPS. A decrease by half of the EOL-CR could lead to a substantial rise of this amount to almost 2 Gt by 2050. Similar results are observed in the NZE scenario, which is explained by the time-delay impact of the lifespan on endof-life flows. The decommissioning scenarios will therefore have significant effects on the possibilities of recycling.

#### 4.3. Associated energy and CO<sub>2</sub> emissions

The energy consumed to produce the materials increased significantly between 1950 and 2019. A domination of steel is observed in the historical and future results. It raised from 0.5 EJ to more than 1 EJ between 1950 and 2019, which represents about 0.15% of the world's primary energy demand in 2019. The energy requirements experience a slight drop to 0.96 EJ in 2050 in the STEPS and to only 0.08 EJ in the NZE scenario in 2050 (Fig. 9). This represents 0.1% of global final energy consumption in 2050 in the STEPS scenario and 0.02% in the NZE. A substantial rise is observed in the CO<sub>2</sub> emissions of sectoral materials production between 1950 and 2019. It evolves from 50 MtCO<sub>2</sub> to more than 130 MtCO<sub>2</sub>, i.e. 0.4% of global emissions in 2019. They first peaked at more than 200 MtCO<sub>2</sub> in 1973, before decreasing sharply in



Figure 7: Historical and prospective stocks of structural materials in the NZE (left column) and STEPS (right column) scenarios.



**Figure 8:** Obsolete steel stock in the hydrocarbon supply chain from 1950 to 2050 in the STEPS.

the 1980s. The trajectories observed in the scenarios allow for a slow decrease to 86  $MtCO_2$  in 2050 in the STEPS scenario and a significant drop to 25  $MtCO_2$  in the NZE scenario (composed only of recycling-related emissions, Fig. 9).

#### 4.4. Sensitivity analysis

The local sensitivity analysis focuses on the mean lifetime value, for which variations of  $\pm 20\%$  were computed. The results show that the materials inflows vary between -11% and +14%, and the outflows changes range from -14% to +19%. Higher impacts are observed in the outflows than on the inflows, as well as in the STEPS than in the NZE scenario.

#### 5. Discussion

## 5.1. Drivers of demand and materials intensities in the fossils supply sector

The bulk materials computed in this study display different drivers. It was found that the midstream and downstream segments are the main drivers of steel, copper and aluminum demand. Conversely, the upstream segment constitutes the main consumer of concrete, ahead of the midstream activities. Those global findings can be extrapolated to national oil, gas, and coal industries. Indeed, a segmentation of countries depending on their profile (producer and/or consumer, oil, gas and/or coal) allows to estimate their materials demand pattern. A mainly consumer country is assumed to have a developed upstream segment on its domestic territory and thus a more significant share of concrete in the material demand than the global share depicted in this study. However, detailed sector data are required to accurately assess national materials demands. Sector materials intensities for the global fossils supply chain were estimated in this study (Fig. 6), and further intensities by segments were assessed (Figures S9 to S12 in the Supplementary Information). It allows for a first order evaluation of the materials consumed

per EJ of fossils produced and can be useful in future topdown materials modeling studies.

## 5.2. What potential for fossil fuels structures to provide materials for the energy transition?

The estimated amount of materials consumed in the hydrocarbon supply chain remains low compared to the global demand-with about 5% of the steel consumption, 1.7% of the copper and smaller share for aluminum and concrete-and to others sectors. A comparison of material intensities, annual demand, cumulative demand and in-use stock with previous studies is provided in Table 2 for both the power infrastructures and the fossil fuels supply chain. The renewable and fossil primary energy supply in scenarios considered in the studies are summarized in Table 3. The current steel demand and in-use stock estimated in this study are higher than in the power infrastructures computed by Deetman et al. [38], while concrete, copper and aluminum needs remain well below. A similar trend is observed for the in-use stocks, and an estimation of the base materials need for the fossil fuels infrastructures in SSP2 BL and SSP2 450 are provided in Table 2 for comparison basis with the results for renewable energies, grid and storage in Deetman et al. [38]. Conversely to copper, aluminum and concrete, the steel in-use stock in the fossil fuels supply chain remains twofold the amount in the renewables and power infrastructures in 2050.

Only the recycling of steel from end-of-life or decommissioned fossil fuels structures could potentially provide a significant contribution to the construction of low-carbon infrastructures. However, it will depend on both future fossil fuel and renewable supply. Relying on data scenarios and sectoral material intensities (Table 2), an estimation of the available recycled steel is provided in Table 3 and hereafter compared with the cumulative demand for the renewable technologies. An EOL-RR of 0.8 was assumed, and the actual available amounts will ultimately depend on the evolution of the collection rate, as depicted in Section 4.2.4. We observe that the increase of fossil fuels demand in the STEPS, SSP2 BL and 450 hinders any secondary steel use in the construction of low-carbon infrastructures. Conversely, moderate reduction of fossil consumption in the Bluemap, 2D and B2D scenarios could allow to partially supply the cumulative steel need of the power technologies<sup>13</sup>. Finally, significant amounts of steel could be recycled in the WWF-Ecofys, NZE and GO scenarios, which would potentially meet the cumulative steel demand of the renewable and power storage infrastructures in the WWF-Ecofys scenario. It would only represent about 17% of the need in the GO scenario, which considers a 100% renewable energy mix. Through intensive recycling of decommissioned infrastructures, the accumulated steel stock in the fossil fuels supply chain could therefore represent an achievable means to reduce the energy and environmental toll of material use for the energy transition.

 $<sup>^{13}\</sup>mbox{Careful}$  attention must be paid to the technological scope of each study.



(a) Embodied energy

(b) Embodied CO<sub>2</sub> emissions

Figure 9: Embodied energy and  $CO_2$  emissions of materials supply in the STEPS and NZE scenario.

References	Steel ( <i>iron</i> )	Concrete	Copper	Aluminum	Years	Technologies	Scenarios	
Sectoral material intensities (Mt/EJ)								
Deetman et al. [38]	5.9	54	0.4	1.5	2015	PG+G+S	Historical	
This study	4	2	0.02	0.002	2015	HU+HM+HD	Historical	
This study	3.8	2.5	0.023	0.006	2015	Oil	Historical	
This study	9.7	3.8	0.038	0	2015	Gas	Historical	
Demand (Mt/yr)								
Deetman et al. [38]	29	184	2.1	7	2015	PG+G+S	Historical	
This study	79	30	0.4	0.04	2015	HU+HM+HD	Historical	
Deetman et al. [38]	58	108	4.2	14	2050	PG+G+S	SSP2 BL	
Deetman et al. [38]	75	227	4.9	16	2050	PG+G+S	SSP2 450	
Vidal [9]	50	200	1.7	5.5	2060	R	Bluemap	
Vidal [9]	600	1600	5.5	23	2060	R	GO	
This study	2.3	0.2	0.01	0.005	2050	HU+HM+HD	NZE	
This study	86	34	0.37	0.05	2050	HU+HM+HD	STEPS	
		Cu	mulative d	emand (Mt)				
Vidal et al. [57]	1,000	N/A	40	160	2000-2050	R	Bluemap	
Vidal et al. [57]	12,000	N/A	200	300	2000-2050	R	GO	
Månberger and Stenqvist [96]	N/A	N/A	40-170	N/A	2010-2050	R+T	B2D	
Moreau et al. [59]	840	N/A	57	94	2010-2050	R+S	WWF	
Moreau et al. [59]	1,500	N/A	32	46	2010-2050	R+S	2D	
Moreau et al. [59]	900	N/A	78	130	2010-2050	R+S	IRENA REMAP	
This study	1,000	390	4.6	0.6	2010-2050	HU+HM+HD	NZE	
This study	3,200	1,300	14	1.9	2010-2050	HU+HM+HD	STEPS	
			In-use sto	ck (Mt)				
Deetman et al. [38]	521	4,772	38	132	2015	PG+G+S	Historical	
This study	2,000	1,000	9.3	1.1	2015	HU+HM+HD	Historical	
Deetman et al. [38]	1,456	9,199	98	365	2050	PG+G+S	SSP2 BL	
This study	2,880	1,390	12.6	0.77	2050	HU+HM+HD	SSP2 BL	
Deetman et al. [38]	1,413	8,386	91	319	2050	PG+G+S	SSP2 450	
This study	2,700	1,340	12	0.86	2050	HU+HM+HD	SSP2 450	
Vidal [9]	5,000	1,200	40	120	2060	R	Bluemap	
Vidal [9]	32,000	13,200	140	480	2060	R	GO	
This study	820	350	3.5	0.3	2050	HU+HM+HD	NZE	
This study	2,900	1,250	12.7	1.4	2050	HU+HM+HD	STEPS	

#### Table 2

Summary of the materials intensities, demand, cumulative demand and in-use stock consumed in the power generation, storage and grid infrastructures and comparison with the hydrocarbon supply chain. PG = power generation, R = renewables, G = power grid, S = power storage, T = electric transportation, HU = hydrocarbons upstream, HM = hydrocarbons midstream, HD = hydrocarbons downstream, GO = scenario by García-Olivares et al. [97]. Value do not sum due to rounding.

Dynamic modeling of global fossil fuel infrastructure and materials needs: overcoming the lack of available data

Scenario	Year	Renewable supply (EJ)	Fossil supply (EJ)	Total TPES (EJ)	Estimated available recycled steel (Mt)	Reference
Historical	2015	23	370	474	N/A	IEA [98]
SSP2 BL	2050	84	694	842	0	Riahi et al. [99]
SSP2 450	2050	98	590	771	0	Riahi et al. [99]
Bluemap	2050	125	297	674	510	IEA [100]
WWF-Ecofys	2050	144	13	262	1,500	Singer et al. [101]
2D	2050	152	269	633	470	IEA [102]
B2D	2050	172	192	581	830	IEA [102]
STEP	2050	193	490	742	0	IEA [98]
IRENA REMAP	2050	222	166	494	N/A	IRENA [103]
NZE	2050	258	110	550	1,000	IEA [87]
GO	2050	389	0	389	2,000	García-Olivares et al. [97]

#### Table 3

Primary renewable, fossil and total energy supply in 2050 in energy transition scenarios. Bioenergy is not accounted for in the renewable supply. The available recycled steel is estimated from the steel in-use stock in 2015 and the scenarios fossil supply in 2050, assuming a 0.8 EOL-RR. GO = scenario by García-Olivares et al. [97].

# 5.3. The need to conduct ambitious decommissioning strategies

The significant potential of recycling materials-and especially steel-in end-of-life fossil fuels structures was identified in section 5.2. We estimated that the current in-use stock of steel could provide large amount of secondary materials to build the future renewable technologies, but that the future EOL-RR could substantially impact recycled materials availability. Two main elements could affect the EOL-RR. First, most of the copper of the fossil fuels supply chain is consumed in steel alloys (e.g. stainless steel, 7% nickel steel and 36% nickel steel in LNG infrastructures), for which down-cycling<sup>14</sup> is frequently observed. Hayatama et al. [104] and Ciacci et al. [105] outlined the losses of copper in alloys recycling and changes in functionality of complex recycled steel alloys. Second, partial or total phasing-out of fossil fuels will require decommissioning policies. Many strategies can be adopted depending on the technology (e.g. removal, partial removal, left in place, reefing or left in place and repurposed) and will impact the EOL-RR. The removal process of the infrastructures could generate substantial energy consumption and environmental impacts [106], whichif recycled-could be compared to the savings from avoiding the extraction of primary materials at a global scale in future studies.

Decommissioning remains decisive to avoid both indirect environmental impacts (e.g. preventing primary material production) and direct environmental impacts<sup>15</sup> (e.g. GHG emissions of abandoned wells [109, 110, 111]<sup>16</sup>). Ambitious decommissioning policies could allow to enhance material recycling and reuse<sup>17</sup>. However, some processes will require material demand (e.g. cement plugs of oil and gas wells), a part of the estimated in-use stock will inevitably be left in place (e.g. the materials embedded in the wellbore [113]) and prohibitive collection price could hamper material recovery<sup>18</sup>. The EOL-CR will ultimately be impacted by the primary material prices and the local decommissioning policies. At a global scale, decommissioning remains at its early stage, and while some countries pave the way for ambitious policies [95, 115], other display nascent strategies [116]. However, the climate emergency urges to take rapid action to reduce the environmental toll of human activities, and ambitious decommissioning strategies could provide an additional way to increase materials recycling and reuse.

#### 6. Robustness and limitations

## 6.1. Robustness criteria and local sensitivity analysis

The dynamic modeling of the hydrocarbon production and supply infrastructure stock proposed in this study represents a simple global approach to the sector, focusing on the main infrastructures. This model needs to be robust and transparent, and several robustness criteria have been identified. First, the stock modeling results are satisfactory

<sup>&</sup>lt;sup>14</sup>Down-cycling is defined as the recycling resulting in a lower quality of materials, and therefore a loss of functionalities.

<sup>&</sup>lt;sup>15</sup>The materials and environmental opportunities of decommissioning have been emphasized [107, 108].

<sup>&</sup>lt;sup>16</sup>Significant methane emissions have also been identified during the operational phase [112]. The study shows that large methane emissions during maintenance or failures of oil and gas equipments could represent between 8% and 12% of the sector global methane emissions.

<sup>&</sup>lt;sup>17</sup>Davies and Hastings [111] estimated that a 16,000 tons steel jacket left in place and repurposed could retain 55,040 tCO<sub>2</sub>eq in GHG emissions.

<sup>&</sup>lt;sup>18</sup>Raimi et al. [114] estimated median prices for wells plugging and surface reclamation to \$76,000 in the US, with significant disparities.

for nine of the eleven infrastructure types modeled from energy production. A first category of modeled infrastructure stocks displays averaged errors of less than 10% (wells, gas and oil product pipelines, LNG liquefaction and regasification plants, LNG storage facilities and refineries) and a higher value are observed for coal carriers (11%), oil tankers (12%) and LNG carriers (16%). Second, the global approach developed in this study allows to assess the evolution of hydrocarbon supply chain from a minimal panel of characteristics of the hydrocarbon production and supply sector. This approach, although minimal, does allow for the analysis of the evolution of infrastructure stocks according to assumptions such as the progression of LNG or nonconventional hydrocarbons.

The sensitivity analysis conducted in this study shows moderate effects of lifetime changes. However, as previously described by Dong et al. [117] and Yin et al. [118], the time delay induced by the lifetime carries growing effects over time. A better knowledge of technological and regional values of lifetime over time is needed to prevent substantial variations in prospective dynamic modeling. It can also be noticed linear impacts of the material intensities on stocks and flows of the model. Global homogeneous data were assumed in this analysis, and a better understanding of the regional dynamics could increase the accuracy of the results. Finally, local sensitivity analysis only describes variations in a given point in space. This methodology therefore carries limitations [119], and further studies should develop global sensitivity analysis on more parameters and their possible interactions [120].

#### 6.2. Methodology-related limitations

The hydrocarbon production and supply sector are computed independently of the other sectors, and the materials recycled are consumed in the same sector. This approach does not represent the actual recycling industry, and only a multi-sector study considering recycling as a sector would estimate accurate flows of recycled metals between the consumer sectors. In addition, the materials consumed for well closure is not considered [113], as well as the numerous decommissioning strategies (Section 5.3). The choice of a physical approach to the flows and stocks of materials also carries limitations on the effectiveness of the model in terms of operational reality of the sector. For example, an increase in concrete demand is observed in the late 2040s in the NZE scenario, despite a context of significant slowdown in hydrocarbons production. It is explained by end-of-life renewal of some infrastructures. Assumptions of life extension, well closure for profitability considerations and other sectoral and financial realities could be considered in further studies.

#### 6.3. Data-related limitations

Most of current infrastructures data on the fossil fuels supply chain are difficult to access. To overcome this problem, the sector's players call on the services of economic intelligence companies (IHS Markit, Rystad Energy, Wood Mackenzie, etc.) which collect confidential information (obtained in an undisclosed manner) and provide data, tools and analyses in return of a high price. Conversely, this study relies on free and online available dataset describing the evolution of the global fossil fuels supply chain over time. The dataset displayed in the Supplementary Information aims at providing global data for future studies. Further free regional data on fossil fuel infrastructures were gathered in other studies, for instance Byrnes [121]. In the present study, the data require extensive sources research, remain scarce and suffer from a lack of sources comparison and from numerous gaps, which hamper any extrapolation over time. For instance, access to reliable data over time and consistent over the geographical perimeter remains a limitation to this study. The selected data were qualified according to their geographical scope, technological scope and year of the study. The data collected do not allow a complete understanding of the complexity of the hydrocarbon production sector, due to the great variety of systems used from exploration to distribution<sup>19</sup>. Detailed data on all these technologies and their associated material intensities are not yet available, and sub-segment scale data were therefore considered<sup>20</sup>. Thus, the results displayed in the study are likely an underestimate of the actual material demand. In addition, most of the data are based on European or North American devices, which could create high uncertainties in the results. This is explained by the difference in both the global hydrocarbons production and consumption patterns<sup>21</sup>.

Furthermore, our analysis relies on oil production scenarios which can be thrown into question. Indeed, the International Energy Agency tends to assume parameters which proved to result in too optimistic projections in the past [124, 125]. It consequently neglects the possibility to suffer from a production crunch in the coming years [126] or to experience a peak in the production of all oil liquids in the mid 2030s now envisioned by an important number of actors [127]. While the issue of credible long-term oil scenarios is critical at times of the resurgence of the 'peak oil' debate [128], it seems not to be taken into account by the IEA. We acknowledge the fact that projecting oil demand and supply is an intricate and complex task. Yet, we feel that the projections we used carry great uncertainty, and that they should be treated with a grain of salt.

#### 7. Conclusions

This study models the evolution of the world's fossil fuel infrastructures and estimates dynamic embedded materials, energy, and  $CO_2$  emissions from 1950 to 2050. To do so, it

<sup>&</sup>lt;sup>19</sup>A detailed list of equipment used in the offshore production of hydrocarbons is provided by Rosneft.

<sup>&</sup>lt;sup>20</sup>For example in the exploration sub-segment, see Supplementary Information.

<sup>&</sup>lt;sup>21</sup>Oil and gas drilling technologies are different from country to country, with conventional hydrocarbons requiring fewer wells drilled than unconventional hydrocarbons for an equivalent level of production. There is therefore a lack of correlation between the amount of oil produced and the number of wells drilled at the regional level, with OPEC, for example, accounting for nearly 32% of global oil production and 15% of natural gas production between 1980 and 2019 for only 2.8% of wells completed over the same period [122, 123].

firstly collects and centralizes free data from a large number of sources and thus avoids using expensive datasets from energy intelligence firms (McKenzie, IHS Markit, Rystad Energy, etc.). Without claiming to provide the same quality and disaggregation of data, our comprehensive dataset can be used by other researchers to understand the role of hydrocarbon infrastructures in the energy transition. Secondly, it develops and applies a dynamic material flow analysis (MFA).

We find that (i) the material intensities of oil, gas and coal have stagnated for more than 30 years; (ii) gas is the main driver of current and future material consumption and (iii) recycled steel from decommissioned fossil fuels infrastructures could meet the cumulative need of future lowcarbon technologies and reduce its energy and environmental toll.

We furthermore highlight that the regional decommissioning strategies will significantly affect the potential of material recycling and reuse. In this context, ambitious decommissioning strategies could drive a symbolic move to build future renewable technologies from past fossil fuel structures.

#### **CRediT** authorship contribution statement

**Hugo Le Boulzec**: Conceptualization, Methodology, Software, Data Curation, Writing - Original Draft, Writing - Review & Editing. **Louis Delannoy**: Conceptualization, Writing- Original draft preparation, Writing - Review & Editing. **Baptiste Andrieu**: Visualization, Writing - Review & Editing. **François Verzier**: Software. **Olivier Vidal**: Writing - Review & Editing, Supervision, Funding acquisition. **Sandrine Mathy**: Writing - Review & Editing, Supervision.

#### **Declaration of competing interests**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper

#### Acknowledgments

We would like to thank the two anonymous reviewers for their comments that helped improving our manuscript. We thank Roger Bentley, Hugo Duterne and Jean Laherrère and Christophe Roncato for reviewing this manuscript, as well as Daniel Monfort-Climent, Faustine Laurent, Jacques Villeneuve from the French Geological Survey (BRGM) and Alain Geldron and Jean-Michel Parrouffe from the French Environmental Agency (ADEME) for fruitful discussions.

#### Funding

This work was supported by the project SURFER financed by the French Environment and Energy Management Agency (ADEME), the French National Institute for Research in Digital Science and Technology (INRIA) and the grant ANR-21-CE03-0012 SCARCYCLET without any involvement in the conduct of the research and the preparation of the article.

#### References

- Krausmann F, Gingrich S, Eisenmenger N, Erb KH, Haberl H, Fischer-Kowalski M. Growth in global materials use, GDP and population during the 20th century. Ecological Economics. 2009 Aug;68(10):2696-705. Available from: https://linkinghub.elsevier. com/retrieve/pii/S092180090902158.
- [2] Fischer-Kowalski M. Analyzing sustainability transitions as a shift between socio-metabolic regimes. Environmental Innovation and Societal Transitions. 2011 Jun;1(1):152-9. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S2210422411000153.
- [3] Krausmann F, Schaffartzik A, Mayer A, Eisenmenger N, Gingrich S, Haberl H, et al. Long-Term Trends in Global Material and Energy Use. In: Haberl H, Fischer-Kowalski M, Krausmann F, Winiwarter V, editors. Social Ecology. Cham: Springer International Publishing; 2016. p. 199-216. Available from: http: //link.springer.com/10.1007/978-3-319-33326-7\_8.
- [4] Hall CAS, Lambert JG, Balogh SB. EROI of different fuels and the implications for society. Energy Policy. 2014 Jan;64:141-52. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S0301421513003856.
- [5] Jackson T. The Post-growth Challenge: Secular Stagnation, Inequality and the Limits to Growth. Ecological Economics. 2019 Feb;156:236-46. Available from: https://linkinghub.elsevier.com/ retrieve/pii/S0921800918309455.
- [6] IPCC. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change; 2021.
- [7] Bihouix P. Quel futur pour les métaux : raréfaction des métaux : un nouveau défi pour la société. Les Ulis: EDP sciences; 2010.
- [8] Vidal O, Goffé B, Arndt N. Metals for a low-carbon society. Nature Geoscience. 2013 oct;6(11):894-6.
- [9] Vidal O. Mineral resources and energy: future stakes in energy transition. London: ISTE Press Ltd; 2018.
- [10] Ali SH, Giurco D, Arndt N, Nickless E, Brown G, Demetriades A, et al. Mineral supply for sustainable development requires resource governance. Nature. 2017 mar;543(7645):367-72.
- [11] Bleicher A. The material basis of energy transitions. London San Diego, CA: Academic Press; 2020.
- [12] Fizaine F. Les métaux rares : opportunité ou menace ? Enjeux et perspectives associés à la transition énergétique. Paris: Editions Technip; 2021.
- [13] Graedel TE, Bertram M, Fuse K, Gordon RB, Lifset R, Rechberger H, et al. The contemporary

European copper cycle: The characterization of technological copper cycles. Ecological Economics. 2002 Aug;42(1-2):9-26. Available from: https://linkinghub.elsevier.com/retrieve/ pii/S0921800902001015.

- [14] Spatari S, Bertram M, Fuse K, Graedel TE, Rechberger H. The contemporary European copper cycle: 1 year stocks and flows. Ecological Economics. 2002 Aug;42(1-2):27-42. Available from: https://linkinghub.elsevier.com/retrieve/ pii/S0921800902001039.
- [15] Kapur A, Bertram M, Spatari S, Fuse K, Graedel TE. The contemporary copper cycle of Asia. Journal of Material Cycles and Waste Management. 2003 Aug;5(2):143-56. Available from: http://link. springer.com/10.1007/s10163-003-0096-4.
- [16] Vexler D, Bertram M, Kapur A, Spatari S, Graedel TE. The contemporary Latin American and Caribbean copper cycle: 1 year stocks and flows. Resources, Conservation and Recycling. 2004 Apr;41(1):23-46. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S0921344903001198.
- [17] International Copper Study Group. ICSG Statistical Database; 2021. Available from: http://www.icsg. org/.
- [18] World Bureau of Metal Statistics. Metals database; 2021. Available from: https://world-bureau.co.uk/.
- [19] Müller D. Stock dynamics for forecasting material flows—Case study for housing in The Netherlands. Ecological Economics. 2006 Aug;59(1):142-56. Available from: https://linkinghub.elsevier.com/retrieve/pii/S092180090500460X.
- [20] Hashimoto S, Tanikawa H, Moriguchi Y. Framework for estimating potential wastes and secondary resources accumulated within an economy A case study of construction minerals in Japan. Waste Management. 2009 Nov;29(11):2859-66. Available from: https://linkinghub.elsevier.com/retrieve/pii/S0956053X0900227X.
- [21] Arora M, Raspall F, Cheah L, Silva A. Residential building material stocks and componentlevel circularity: The case of Singapore. Journal of Cleaner Production. 2019 Apr;216:239-48. Available from: https://linkinghub.elsevier.com/ retrieve/pii/S095965261930246X.
- [22] Wiedenhofer D, Steinberger JK, Eisenmenger N, Haas W. Maintenance and Expansion: Modeling Material Stocks and Flows for Residential Buildings and Transportation Networks in the EU25: Stocks and Flows in the EU25. Journal of Industrial Ecology. 2015 Aug;19(4):538-51. Available from: http://doi. wiley.com/10.1111/jiec.12216.
- [23] Hu M, Pauliuk S, Wang T, Huppes G, van der Voet E, Müller DB. Iron and steel in Chinese residential buildings: A dynamic analysis. Resources,

Conservation and Recycling. 2010 Jul;54(9):591-600. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0921344909002407.

- [24] Deetman S, Marinova S, van der Voet E, van Vuuren DP, Edelenbosch O, Heijungs R. Modelling global material stocks and flows for residential and service sector buildings towards 2050. Journal of Cleaner Production. 2020 feb;245:118658.
- [25] Marinova S, Deetman S, van der Voet E, Daioglou V. Global construction materials database and stock analysis of residential buildings between 1970-2050. Journal of Cleaner Production. 2020 feb;247:119146.
- [26] Watari T, McLellan BC, Giurco D, Dominish E, Yamasue E, Nansai K. Total material requirement for the global energy transition to 2050: A focus on transport and electricity. Resources, Conservation and Recycling. 2019 sep;148:91-103.
- [27] Tokimatsu K, Höök M, McLellan B, Wachtmeister H, Murakami S, Yasuoka R, et al. Energy modeling approach to the global energy-mineral nexus: Exploring metal requirements and the well-below 2°C target with 100 percent renewable energy. Applied Energy. 2018 sep;225:1158-75.
- [28] Kleijn R, van der Voet E, Kramer GJ, van Oers L, van der Giesen C. Metal requirements of low-carbon power generation. Energy. 2011 sep;36(9):5640-8.
- [29] Elshkaki A, Graedel TE. Dynamic analysis of the global metals flows and stocks in electricity generation technologies. Journal of Cleaner Production. 2013 nov;59:260-73.
- [30] IEA. The Role of Critical Minerals in Clean Energy Transitions. IEA. 2020. Available from: https://iea.blob.core.windows.net/ assets/24d5dfbb-a77a-4647-abcc-667867207f74/ TheRoleofCriticalMineralsinCleanEnergyTransitions. pdf.
- [31] Li F, Ye Z, Xiao X, Xu J, Liu G. Material stocks and flows of power infrastructure development in China. Resources, Conservation and Recycling. 2020 Sep;160:104906. Available from: https://linkinghub.elsevier.com/retrieve/ pii/S092134492030224X.
- [32] de Koning A, Kleijn R, Huppes G, Sprecher B, van Engelen G, Tukker A. Metal supply constraints for a low-carbon economy? Resources, Conservation and Recycling. 2018 feb;129:202-8.
- [33] Hertwich EG, Gibon T, Bouman EA, Arvesen A, Suh S, Heath GA, et al. Integrated life-cycle assessment of electricity-supply scenarios confirms global environmental benefit of low-carbon technologies. Proceedings of the National Academy of Sciences. 2014 oct;112(20):6277-82.
- [34] Elshkaki A, Shen L. Energy-material nexus: The impacts of national and international energy scenarios on critical metals use in China up to 2050 and their

global implications. Energy. 2019 Aug;180:903-17. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0360544219310382.

- [35] Giurco D, Dominish E, Florin N, Watari T, McLellan B. Requirements for Minerals and Metals for 100% Renewable Scenarios. In: Achieving the Paris Climate Agreement Goals. Springer International Publishing; 2019. p. 437-57.
- [36] National Research Council. Minerals, Critical Minerals, and the U.S. Economy. National Academies Press; 2008.
- [37] Månberger A, Stenqvist B. Global metal flows in the renewable energy transition: Exploring the effects of substitutes, technological mix and development. Energy Policy. 2018 aug;119:226-41.
- [38] Deetman S, de Boer HS, Engelenburg MV, van der Voet E, van Vuuren DP. Projected material requirements for the global electricity infrastructure – generation, transmission and storage. Resources, Conservation and Recycling. 2021 jan;164:105200.
- [39] Elshkaki A. Materials, energy, water, and emissions nexus impacts on the future contribution of PV solar technologies to global energy scenarios. Scientific Reports. 2019 dec;9(1).
- [40] Davidsson S, Höök M. Material requirements and availability for multi-terawatt deployment of photovoltaics. Energy Policy. 2017 sep;108:574-82.
- [41] Nassar NT, Wilburn DR, Goonan TG. Byproduct metal requirements for U.S. wind and solar photovoltaic electricity generation up to the year 2040 under various Clean Power Plan scenarios. Applied Energy. 2016 dec;183:1209-26.
- [42] Teubler J, Kiefer S, Liedtke C. Metals for Fuels? The Raw Material Shift by Energy-Efficient Transport Systems in Europe. Resources. 2018 aug;7(3):49.
- [43] Shammugam S, Gervais E, Schlegl T, Rathgeber A. Raw metal needs and supply risks for the development of wind energy in Germany until 2050. Journal of Cleaner Production. 2019 jun;221:738-52.
- [44] Beylot A, Guyonnet D, Muller S, Vaxelaire S, Villeneuve J. Mineral raw material requirements and associated climate-change impacts of the French energy transition by 2050. Journal of Cleaner Production. 2019 jan;208:1198-205.
- [45] Ren K, Tang X, Wang P, Willerström J, Höök M. Bridging energy and metal sustainability: Insights from China's wind power development up to 2050. Energy. 2021 Jul;227:120524. Available from: https://linkinghub.elsevier.com/ retrieve/pii/S0360544221007738.
- [46] Habib K, Wenzel H. Exploring rare earths supply constraints for the emerging clean energy technologies and the role of recycling. Journal of Cleaner Production. 2014 dec;84:348-59.

- [47] Kavlak G, McNerney J, Jaffe RL, Trancik JE. Metal production requirements for rapid photovoltaics deployment. Energy & Environmental Science. 2015;8(6):1651-9.
- [48] Grandell L, Lehtilä A, Kivinen M, Koljonen T, Kihlman S, Lauri LS. Role of critical metals in the future markets of clean energy technologies. Renewable Energy. 2016 sep;95:53-62.
- [49] Knoeri C, Wäger PA, Stamp A, Althaus HJ, Weil M. Towards a dynamic assessment of raw materials criticality: Linking agent-based demand — With material flow supply modelling approaches. Science of The Total Environment. 2013 sep;461-462:808-12.
- [50] Moss RL, Tzimas E, Kara H, Willis P, Kooroshy J. The potential risks from metals bottlenecks to the deployment of Strategic Energy Technologies. Energy Policy. 2013 apr;55:556-64.
- [51] Goe M, Gaustad G. Identifying critical materials for photovoltaics in the US: A multi-metric approach. Applied Energy. 2014 jun;123:387-96.
- [52] Roelich K, Dawson DA, Purnell P, Knoeri C, Revell R, Busch J, et al. Assessing the dynamic material criticality of infrastructure transitions: A case of low carbon electricity. Applied Energy. 2014 jun;123:378-86.
- [53] Jin Y, Kim J, Guillaume B. Review of critical material studies. Resources, Conservation and Recycling. 2016 oct;113:77-87.
- [54] Valero A, Valero A, Calvo G, Ortego A. Material bottlenecks in the future development of green technologies. Renewable and Sustainable Energy Reviews. 2018 oct;93:178-200.
- [55] Lee J, Bazilian M, Sovacool B, Hund K, Jowitt SM, Nguyen TP, et al. Reviewing the material and metal security of low-carbon energy transitions. Renewable and Sustainable Energy Reviews. 2020 may;124:109789.
- [56] Watari T, McLellan B, Ogata S, Tezuka T. Analysis of Potential for Critical Metal Resource Constraints in the International Energy Agency's Long-Term Low-Carbon Energy Scenarios. Minerals. 2018 apr;8(4):156.
- [57] Vidal O, Rostom F, François C, Giraud Global Trends in Metal Consumption G. and Supply: The Raw Material-Energy Nexus. Elements. 2017 Oct;13(5):319-24. Available from: https://pubs.geoscienceworld. org/msa/elements/article/13/5/319/522742/

Global-Trends-in-Metal-Consumption-and-Supply-The.

- [58] Vidal O, Le Boulzec H, Andrieu B, Verzier F. Modelling the Demand and Access of Mineral Resources in a Changing World. Sustainability. 2021 Dec;14(1):11. Available from: https://www.mdpi.com/ 2071-1050/14/1/11.
- [59] Moreau V, Reis PD, Vuille F. Enough Metals? Resource Constraints to Supply a Fully Renewable Energy System. Resources. 2019 jan;8(1):29.

- [60] Pauliuk S, Arvesen A, Stadler K, Hertwich EG. Industrial ecology in integrated assessment models. Nature Climate Change. 2017 Jan;7(1):13-20. Available from: http://www.nature.com/articles/nclimate3148.
- [61] Hache E, Seck GS, Simoen M, Bonnet C, Carcanague S. Critical raw materials and transportation sector electrification: A detailed bottom-up analysis in world transport. Applied Energy. 2019 apr;240:6-25.
- [62] Hache E, Simoën M, Seck GS, Bonnet C, Jabberi A, Carcanague S. The impact of future power generation on cement demand: An international and regional assessment based on climate scenarios. International Economics. 2020 oct;163:114-33.
- [63] Seck GS, Hache E, Bonnet C, Simoën M, Carcanague S. Copper at the crossroads: Assessment of the interactions between low-carbon energy transition and supply limitations. Resources, Conservation and Recycling. 2020 dec;163:105072.
- [64] Sverdrup HU, Olafsdottir AH. System Dynamics Modelling of the Global Extraction, Supply, Price, Reserves, Resources and Environmental Losses of Mercury. Water, Air, & Soil Pollution. 2020 aug;231(8).
- [65] Olafsdottir AH, Sverdrup HU. Assessing the Past and Future Sustainability of Global Helium Resources, Extraction, Supply and Use, Using the Integrated Assessment Model WORLD7. Biophysical Economics and Sustainability. 2020 may;5(2).
- [66] Olafsdottir AH, Sverdrup HU. Modelling Global Nickel Mining, Supply, Recycling, Stocks-in-Use and Price Under Different Resources and Demand Assumptions for 1850–2200. Mining, Metallurgy & Exploration. 2021 jan;38(2):819-40.
- [67] Capellán-Pérez I, de Castro C, González LJM. Dynamic Energy Return on Energy Investment (EROI) and material requirements in scenarios of global transition to renewable energies. Energy Strategy Reviews. 2019 nov;26:100399.
- [68] Wang T, Müller DB, Hashimoto S. The Ferrous Find: Counting Iron and Steel Stocks in China's Economy: Counting Iron and Steel Stocks in China's Economy. Journal of Industrial Ecology. 2015 Oct;19(5):877-89. Available from: https://onlinelibrary.wiley. com/doi/10.1111/jiec.12319.
- [69] Gutiérrez R, Gutiérrez-Sánchez R, Nafidi A. Modelling and forecasting vehicle stocks using the trends of stochastic Gompertz diffusion models: The case of Spain. Applied Stochastic Models in Business and Industry. 2009 May;25(3):385-405. Available from: https://onlinelibrary.wiley.com/doi/10.1002/ asmb.754.
- [70] Huang T, Shi F, Tanikawa H, Fei J, Han J. Materials demand and environmental impact of buildings construction and demolition in China based on dynamic material flow analysis. Resources, Conservation and Recycling. 2013 Mar;72:91-101.

Available from: https://linkinghub.elsevier.com/ retrieve/pii/S0921344912002273.

- [71] BP. Statistical Review of World Energy. BP; 2021. 70th edition. Available from: https://www.bp.com/content/dam/ bp/business-sites/en/global/corporate/ pdfs/energy-economics/statistical-review/ bp-stats-review-2021-full-report.pdf.
- [72] IEA. World Energy Outlook 2019. IEA; 2019.
- [73] USGS. Mineral Commodity Summaries. USGS; 2021.
- [74] Bataille C. Low and zero emissions in the steel and cement industries. Barriers, technologies and policies. OECD; 2019.
- [75] Sverdrup HU, Ragnarsdottir KV, Koca D. On modelling the global copper mining rates, market supply, copper price and the end of copper reserves. Resources, Conservation and Recycling. 2014 Jun;87:158-74. Available from: http://linkinghub. elsevier.com/retrieve/pii/S0921344914000652.
- [76] Sverdrup HU, Olafsdottir AH. Assessing the Long-Term Global Sustainability of the Production and Supply for Stainless Steel. BioPhysical Economics and Resource Quality. 2019 Jun;4(2):8. Available from: http://link.springer.com/10.1007/ s41247-019-0056-9.
- [77] Glöser S, Soulier M, Tercero Espinoza LA. Dynamic Analysis of Global Copper Flows. Global Stocks, Postconsumer Material Flows, Recycling Indicators, and Uncertainty Evaluation. Environmental Science & Technology. 2013 Jun;47(12):6564-72. Available from: http://pubs.acs.org/doi/abs/10. 1021/es400069b.
- [78] Graedel TE, Allwood J, Birat JP, Buchert M, Hagelüken C, Reck BK, et al. What Do We Know About Metal Recycling Rates? Journal of Industrial Ecology. 2011 Jun;15(3):355-66. Available from: https://onlinelibrary.wiley.com/doi/10.1111/ j.1530-9290.2011.00342.x.
- [79] Monfort-Climent D, Laurent F, Le Boulzec H, Lefevbre G, Rodriguez B, Muller S, et al. Inventaires des besoins en matières, énergie, eau et sols des technologies de la transition énergétique, Projet SURFER. ADEME, ISTerre, BRGM; 2021.
- [80] Court V. Energy, EROI, and Economic Growth in a Long-Term Perspective [Ph.D. thesis]. Paris Nanterre; 2016. Available from: https://bdr.parisnanterre.fr/theses/internet/ 2016PA100109/2016PA100109.pdf.
- [81] Norgate TE, Jahanshahi S, Rankin WJ. Assessing the environmental impact of metal production processes. Journal of Cleaner Production. 2007 Jan;15(8-9):838-48. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0959652606002320.
- [82] Birat JP, Chiappini M, Ryman C, Riesbeck J. Cooperation and competition among structural materials. Revue de Métallurgie. 2013;110(1):95-129.

Available from: http://www.metallurgical-research. org/10.1051/metal/2013057.

- [83] Gutowski TG, Sahni S, Allwood JM, Ashby MF, Worrell E. The energy required to produce materials: constraints on energyintensity improvements, parameters of demand. Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences. 2013 Jan;371(1986):20120003-3. Available from: http://rsta.royalsocietypublishing.org/cgi/doi/ 10.1098/rsta.2012.0003.
- [84] Vidal O. Modeling the Long-Term Evolution of Primary Production Energy and Metal Prices. In: Mineral Resources Economics 1: Context and Issues. fizaine, f.; galiègue, x. ed. Wiley; 2021.
- [85] Vidal O, Andrieu B, Le Boulzec H, Fizaine F. Modeling the production energy and metal prices from 1900 to 2100. Manuscrit submitted for publication. 2021.
- [86] IEA. World Energy Outlook 2020. IEA; 2020.
- [87] IEA. Net Zero by 2050 A Roadmap for the Global Energy Sector. IEA; 2021.
- [88] IEA. Energy Technology Perspectives 2020. IEA; 2020.
- [89] Pauliuk S, Wang T, Müller DB. Steel all over the world: Estimating in-use stocks of iron for 200 countries. Resources, Conservation and Recycling. 2013 Feb;71:22-30. Available from: http://linkinghub. elsevier.com/retrieve/pii/S0921344912002078.
- [90] Wiedenhofer D, Fishman T, Lauk C, Haas W, Krausmann F. Integrating Material Stock Dynamics Into Economy-Wide Material Flow Accounting: Concepts, Modelling, and Global Application for 1900–2050. Ecological Economics. 2019 Feb;156:121-33. Available from: https://linkinghub.elsevier.com/retrieve/ pii/S0921800918305718.
- [91] Müller DB, Wang T, Duval B. Patterns of Iron Use in Societal Evolution. Environmental Science & Technology. 2011 Jan;45(1):182-8. Available from: https://pubs.acs.org/doi/10.1021/es102273t.
- [92] USGS. U.S. Geological Survey, Mineral Commodity Summaries, January 2021, Cement. USGS; 2021. Available from: https://pubs.usgs.gov/periodicals/ mcs2021/mcs2021-cement.pdf.
- [93] USGS. U.S. Geological Survey, Mineral Commodity Summaries, January 2021, steel. USGS; 2021. Available from: https://pubs.usgs.gov/periodicals/ mcs2021/mcs2021-iron-steel.pdf.
- [94] OGUK. Decommissioning Insight 2020. OGUK; 2020.
- [95] Kaiser MJ. A scenario-based deepwater decommissioning forecast in the U.S. Gulf of Mexico. Journal of Petroleum Science and Engineering. 2018:33.
- [96] Månberger A, Stenqvist B. Global metal flows in the renewable energy transition: Exploring the

effects of substitutes, technological mix and development. Energy Policy. 2018 Aug;119:226-41. Available from: https://linkinghub.elsevier.com/retrieve/pii/S0301421518302726.

- [97] García-Olivares A, Ballabrera-Poy J, García-Ladona E, Turiel A. A global renewable mix with proven technologies and common materials. Energy Policy. 2012 Feb;41:561-74. Available from: http://linkinghub. elsevier.com/retrieve/pii/S0301421511008950.
- [98] IEA. World Energy Outlook 2021. IEA; 2021.
- [99] Riahi K, van Vuuren DP, Kriegler E, Edmonds J, O'Neill BC, Fujimori S, et al. The Shared Socioeconomic Pathways and their energy, land use, and greenhouse gas emissions implications: An overview. Global Environmental Change. 2017 Jan;42:153-68. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0959378016300681.
- [100] IEA. Energy Technology Perspectives 2010. IEA; 2010.
- [101] Singer S, World Wildlife Fund, ECOFYS (Firm) OfMA. The energy report: 100% renewable energy by 2050. Gland: WWF International; 2011. OCLC: 705288182. Available from: http://assets.wwfza.panda.org/downloads/101223\_ energy\_report\_final\_print\_2.pdf.
- [102] IEA. Energy Technology Perspectives 2017. IEA; 2017.
- [103] IRENA. Global Energy Transformation: A Roadmap to 2050. IRENA; 2018.
- [104] Hatayama H, Daigo I, Matsuno Y, Adachi Y. Outlook of the World Steel Cycle Based on the Stock and Flow Dynamics. Environmental Science & Technology. 2010 Aug;44(16):6457-63. Available from: https: //pubs.acs.org/doi/10.1021/es100044n.
- [105] Ciacci L, Vassura I, Passarini F. Urban Mines of Copper: Size and Potential for Recycling in the EU. Resources. 2017 Jan;6(1):6. Available from: http: //www.mdpi.com/2079-9276/6/1/6.
- [106] Tan Y, Li HX, Cheng JCP, Wang J, Jiang B, Song Y, et al. Cost and environmental impact estimation methodology and potential impact factors in offshore oil and gas platform decommissioning: A review. Environmental Impact Assessment Review. 2021 Mar;87:106536. Available from: https://linkinghub. elsevier.com/retrieve/pii/S0195925520308143.
- [107] Ekins P, Vanner R, Firebrace J. Decommissioning of offshore oil and gas facilities: A comparative assessment of different scenarios. Journal of Environmental Management. 2006 Jun;79(4):420-38. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0301479705002847.
- [108] Sommer B, Fowler AM, Macreadie PI, Palandro DA, Aziz AC, Booth DJ. Decommissioning of offshore oil and gas structures – Environmental opportunities and challenges. Science of The Total Environment. 2019 Mar;658:973-81. Available from: https://linkinghub.elsevier.

com/retrieve/pii/S0048969718350551.

- [109] Boothroyd IM, Almond S, Qassim SM, Worrall F, Davies RJ. Fugitive emissions of methane from abandoned, decommissioned oil and gas wells. Science of The Total Environment. 2016 Mar;547:461-9. Available from: https://linkinghub.elsevier.com/ retrieve/pii/S0048969715312535.
- [110] Kang M, Christian S, Celia MA, Mauzerall DL, Bill M, Miller AR, et al. Identification and characterization of high methane-emitting abandoned oil and gas wells. Proceedings of the National Academy of Sciences. 2016 Nov;113(48):13636-41. Available from: https://pnas.org/doi/full/10.1073/pnas.1605913113.
- [111] Davies AJ, Hastings A. Quantifying greenhouse gas emissions from decommissioned oil and gas steel structures: Can current policy meet NetZero goals? Energy Policy. 2022 Jan;160:112717. Available from: https://linkinghub.elsevier.com/ retrieve/pii/S0301421521005826.
- [112] Lauvaux T, Giron C, Mazzolini M, d'Aspremont A, Duren R, Cusworth D, et al. Global Assessment of Oil and Gas Methane Ultra-Emitters. Atmospheric Sciences; 2021. Available from: http://eartharxiv. org/repository/view/2323/.
- [113] Vrålstad T, Saasen A, Fjær E, Øia T, Ytrehus JD, Khalifeh M. Plug & abandonment of offshore wells: Ensuring long-term well integrity and cost-efficiency. Journal of Petroleum Science and Engineering. 2019 Feb;173:478-91. Available from: https://linkinghub.elsevier.com/ retrieve/pii/S0920410518309173.
- [114] Raimi D, Krupnick AJ, Shah JS, Thompson A. Decommissioning Orphaned and Abandoned Oil and Gas Wells: New Estimates and Cost Drivers. Environmental Science & Technology. 2021 Aug;55(15):10224-30. Available from: https://pubs.acs.org/doi/10.1021/acs.est.1c02234.
- [115] OGUK. Decommissioning Insight 2021. London, United Kingdom: OGUK; 2021. Available from: https://oguk.org.uk/wp-content/uploads/2021/ 11/Decommissioning-Insight-2021-OGUK.pdf.
- [116] Melbourne-Thomas J, Hayes KR, Hobday AJ, Little LR, Strzelecki J, Thomson DP, et al. Decommissioning Research Needs for Offshore Oil and Gas Infrastructure in Australia. Frontiers in Marine Science. 2021 Jul;8:711151. Available from: https://www.frontiersin.org/articles/ 10.3389/fmars.2021.711151/full.
- [117] Dong D, Tukker A, Van der Voet E. Modeling copper demand in China up to 2050: A business-as-usual scenario based on dynamic stock and flow analysis. Journal of Industrial Ecology. 2019 Dec;23(6):1363-80. Available from: https://onlinelibrary.wiley. com/doi/10.1111/jiec.12926.

- [118] Yin X, Chen W. Trends and development of steel demand in China: A bottom-up analysis. Resources Policy. 2013 Dec;38(4):407-15. Available from: https://linkinghub.elsevier. com/retrieve/pii/S0301420713000482.
- [119] Saltelli A, editor. Global sensitivity analysis: the primer. Chichester, England ; Hoboken, NJ: John Wiley; 2008. OCLC: ocn180852094.
- [120] Buchner H, Laner D, Rechberger H, Fellner J. Dynamic Material Flow Modeling: An Effort to Calibrate and Validate Aluminum Stocks and Flows in Austria. Environmental Science & Technology. 2015 May;49(9):5546-54. Available from: https://pubs. acs.org/doi/10.1021/acs.est.5b00408.
- [121] Byrnes R. A Global Registry of Fossil Fuels. Fossil Fuel Non-Proliferation Treaty Initiative; 2020.
- [122] Globalshift. Onshore and offshore production series; 2020. Available from: https://gssubsite. globalshift.co.uk/index.html.
- [123] OPEC. World Oil Outlook 2040. OPEC; 2019. Available from: https://www.opec.org/ opec\_web/static\_files\_project/media/downloads/ publications/W00\_2019.pdf.
- [124] Jakobsson K, Söderbergh B, Höök M, Aleklett K. How reasonable are oil production scenarios from public agencies? Energy Policy. 2009 Nov;37(11):4809-18. Available from: https://doi. org/10.1016/j.enpol.2009.06.042.
- [125] Wachtmeister H, Henke P, Höök M. Oil projections in retrospect: Revisions, accuracy and current uncertainty. Applied Energy. 2018 Jun;220:138-53. Available from: https://doi.org/10.1016/j.apenergy.2018.03.013.
- [126] Hacquard P, Simoën M, Hache E. Is the oil industry able to support a world that consumes 105 million barrels of oil per day in 2025? Oil & Gas Science and Technology – Revue d'IFP Energies nouvelles. 2019;74:88. Available from: https://doi.org/10. 2516/ogst/2019061.
- [127] Tupaz JCVC. Inflection point: The future of subcontracting in the petroleum industry. Transportation Research Interdisciplinary Perspectives. 2020 Jul;6:100159. Available from: https://doi.org/10. 1016/j.trip.2020.100159.
- [128] Delannoy L, Longaretti PY, Murphy DJ, Prados E. Peak oil and the low-carbon energy transition: A net-energy perspective. Applied Energy. 2021 Dec;304:117843. Available from: https://doi.org/ 10.1016/j.apenergy.2021.117843.